Author's response to reviews

Title: An overview of ethical frameworks in public health: can they be supportive in the evaluation of programs to prevent overweight?

Authors:

Marieke ten Have (m.tenhave@erasusmc.nl)
Inez D de Beaufort (i.debeaufort@erasusmc.nl)
Johan P Mackenbach (j.mackenbach@erasusmc.nl)
Agnes van der Heide (a.vanderheide@erasusmc.nl)

Version: 4 Date: 20 October 2010

Author's response to reviews:

Dear editorial office,

We have incorporated the comments below. We have placed the tables behind the text. We have also deleted the vertical lines in appendix 2.

"Tables: tables must be created using a Tables tool, and should not be linked files,
figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables."

Yours sincerely,

Marieke ten Have